Author's response to reviews

Title: Haplotype frequencies in a sub-region of chromosome 19q13.3, related to risk and prognosis of cancer, differ dramatically between ethnic groups

Authors:

Mikkel H Schierup (mheide@daimi.au.dk)
Thomas Mailund (mailund@birc.au.dk)
Heng Li (liheng@genomics.org.cn)
Jun Wang (wangj@genomics.org.cn)
Anne Tjønneland (anнет@cancer.dk)
Ulla Vogel (ulbvo@food.dtu.dk)
Lars Bolund (bolund@humgen.au.dk)
Bjørn A Nexø (nexo@humgen.au.dk)

Version: 7 Date: 26 February 2009

Author's response to reviews: see over
The following alterations have been made:

1) We have revised the figures 1 through 5 as requested. Now no figure numbers on the figures themselves. Cropped. Weblinks now all functional.
2) Emails to all authors included on front page
3) Running title removed.
4) Abbreviations section moved before Competing interests
5) Manuscript scrutinized for typos. A few found and corrected.

Bjørn Nexø